

Received 9 February 2023, accepted 20 February 2023, date of publication 24 February 2023, date of current version 31 March 2023. Digital Object Identifier 10.1109/ACCESS.2023.3248958

### **RESEARCH ARTICLE**

## In-Line Distributed Dispatch of Active and Reactive Power Based on ADMM and Consensus Considering Battery Degradation in Microgrids

JOHN BARCO-JIMÉNEZ<sup>(D1,2,6</sup>, GERMAN OBANDO<sup>(D3,</sup> HAROLD R. CHAMORRO<sup>(D4,</sup> (Senior Member, IEEE), ANDRÉS PANTOJA<sup>5</sup>, EDUARDO CAICEDO BRAVO<sup>1</sup>, AND JOSÉ A. AGUADO<sup>(D6,</sup> (Member, IEEE))

<sup>1</sup>PPIEE, Universidad del Valle, Cali 760032, Colombia

<sup>2</sup>Programa de Ingeniería Electrónica, Universidad CESMAG, Pasto 520003, Colombia

<sup>3</sup>Programa de Ingeniería Electrónica, Universidad del Rosario, Bogotá 111711, Colombia

<sup>4</sup>KTH, Royal Institute of Technology, 114 28 Stockholm, Sweden

<sup>5</sup>Departamento de Ingeniería Electrónica, Universidad de Nariño, Ciudadela Universitaria Torobajo, Pasto 520002, Colombia

<sup>6</sup>Departamento de Ingeniería Eléctrica, Universidad de Málaga, Edificio de Ingenierías UMA, 29071 Málaga, Spain

Corresponding author: John Barco-Jiménez (john.barco@correounivalle.edu.co)

The work of Andrés Pantoja was supported in part by the Project Campus Verde, SGR, Colombia, under Grant BPIN 2020000100041. The work of Eduardo Caicedo Bravo was supported in part by The Colombia Scientific Program within the framework of call named Ecosistema Científico, under Contract FP44842-218-2018. The work of José A. Aguado was supported in part by Junta de Andalucia, Spain, under Project Ref: P2001164.

**ABSTRACT** This work presents a distributed in-line strategy to manage an isolated microgrid by optimizing active and reactive power dispatch. The proposed objective function leads to minimize the operation costs and addresses some technical requirements such as diminishing power losses and voltage deviation. Additionally, the strategy deals with temporal multi-scale goals, i.e., robustness to demand disturbances and variation of renewable resources (a short-term objective), and preservation of the health of battery-based storage systems (a long-term objective). The technique uses alternating directions method of multipliers (ADMM), accelerated consensus, and a novel battery degradation model (Quadratic AH-Throughput model). We test the proposed solution in a case study that includes renewable resources and lead-acid and lithium batteries. To obtain the results of the case study, we employ a co-simulation scheme that uses Matlab and DIgSILENT. Finally, the performance of the method is compared with a centralized optimization technique.

**INDEX TERMS** Alternating direction method of multipliers, battery loss capacity, consensus, distributed optimization, optimal power flow, tertiary control.

#### I. INTRODUCTION

Microgrids are widely used to integrate renewable energy resources and energy storage systems operating in a hierarchical scheme. The primary and secondary levels of the control hierarchy provide system stability through active and reactive power management [1]. For this purpose, the use of locally controlled power converters is common. On the other hand, the tertiary level establishes optimal operating points according to certain technical and economic conditions

The associate editor coordinating the review of this manuscript and approving it for publication was Ahmed Mohamed<sup>(D)</sup>.

(e.g., reduction of operating costs or provision of ancillary services) [2]. Tertiary controllers are in charge of highlevel decisions, which implies that they generally have to lead/execute several tasks [3]. Just to name a few, highlevel control is in charge of increasing the efficiency and robustness of the microgrid, facilitating the integration and management of storage systems, increasing the quality of service, and reducing the control effort of the lower levels of the hierarchy [4]. This large number of tasks could be performed efficiently by a distributed strategy. In fact, the design of centralized strategies implies a high computational effort and the need to implement complex communication networks, increasing the deployment cost [5]. Additionally, from the network perspective, the robustness of centralized strategies is reduced because all the control effort is assumed by a single central node. Thus, if this node fails, the entire network fails [6], [7].

In contrast, distributed control strategies solve many of the aforementioned challenges. First, these strategies share the computational load among several controllers that operate with local information. Thus, they require communication networks with fewer links. Second, distributed strategies improve the resilience of the system to failures because these schemes do not depend on a single central authority [8]. Furthermore, they favor plug-and-play operation [6] and facilitate the integration of distributed energy resources such as renewable units. Distributed methods have been effective to solve the dispatch problem in microgrids using different approaches such as gradient descent [9], [10], consensus algorithms [11], [12], [13], and the alternating direction method of multipliers (ADMM) [14]. However, there are limitations that have not been fully addressed. Most of the methods reported in the literature do not consider relevant aspects of real scenarios such as the joint dispatch of active and reactive power, the optimal management of batteries, or the provision of ancillary services. For instance, [15] and [16] propose fast-converging distributed algorithms based on ADMM and the distributed gradient descent method to solve the economic dispatch problem, responding properly to demand disturbances [17]. Nevertheless, the algorithms do not take into account the management of energy storage devices, which are exposed to a high number of charging and discharging events. Consequently, these methods can significantly reduce the batteries' lifetime. In addition, the optimization does not consider the reactive power management of the microgrid. This fact reduces the capability of the strategy to mitigate voltage deviation and power losses.

Other distributed approaches address the problem of the imbalance between supply and demand caused by variations of demand profile and the inherent intermittency of renewable generators [18], [19]. Strategies based on a sliding horizon such as distributed model predictive control (DMPC) have shown to be a promising alternative to mitigate the imbalance. Some authors have complemented these strategies with machine learning tools to manage uncertainty. For example, [19] predicts irradiance using neural networks, with reduction of unsupplied energy for most of the analyzed cases. Although the proposed strategy has a distributed nature, it requires a high processing capacity for large prediction horizons, especially when it is necessary to consider small time intervals to cope disturbances of demand or irradiance. There are also some distributed strategies that consider the participation of storage systems to provide stability by facing variations in demand and generation. These strategies ensure that the state of charge of the batteries is within appropriate ranges at any time. For instance, the authors in [14] and [20] manage storage systems to meet technical and economic goals over a long-term horizon. The main drawback of long-term planning for battery management is that the ability to deal with short-term disturbances is seriously affected. Planning strategies with long horizons generally employ low time-resolution to avoid computational overload, but this fact reduces the strategy ability to cope with rapid changes in supply/demand.

In addition to the described challenges related to storage systems, having an appropriate battery-degradation model is another problem to solve. This model is essential to design efficient policies that take care of the health status of batteries to increase their lifespan [4]. Although there are models to estimate the degradation of batteries of several technologies, many of the ones reported in the literature are not designed to be efficiently integrated into management algorithms [21]. On the one hand, there are models with a high degree of detail that, due to their complexity, are inadequate for making predictions in real time. On the other hand, the simplest models (e.g., control-oriented models) are not accurate enough to estimate the batteries lifespan [22], [23].

This work tackles the challenge of developing a comprehensive optimization problem and a solution method for a tertiary-level control of microgrids. Most of the optimization problems reported in the literature about tertiary control of microgrids are not extensive. Generally, they do not consider some relevant aspects such as joint dispatch of active and reactive power, provision of ancillary services (losses minimization, voltage deviation reduction), inclusion of a strategy for battery health care, consideration of uncertainty, and planning of a distributed operation. In this article, we formulate a tertiary-level optimization strategy considering all the aforementioned issues. Furthermore, we propose a distributed algorithm (based on ADMM and consensus) for solving the optimization problem. The algorithm allows us to perform an in-line operation due to its computational efficiency, which is guaranteed by dividing the scheduling horizon into two periods. This process speeds up the search for a solution by considering a long period horizon to schedule the batteries and a short horizon to respond to demand or electricity generation disturbances.

On the other hand, we have developed a novel battery degradation model (Quadratic AH-Throughput model) that can be included in the formulation of receding horizon strategies without losing the differentiability of the cost function. This characteristic allows us to use efficient algorithms ([24], [25]) to solve the underlying optimization problem at each step of the receding horizon scheme. In addition, it is worth noting that the proposed battery degradation model can be adapted to lithium and lead-acid technologies and combines two influential variables (DOD and cycles) to achieve an accurate state of health estimation. Unfortunately, these desirable features are not included in many models reported in the literature. Summarizing, the main contributions of this paper are:

1) A comprehensive model for the tertiary control level of microgrids, which includes the implementation of a distributed strategy based on ADMM and consensus. This strategy can operate in-line and includes multiple programming horizons allowing to deal with both battery management and system disturbances with a low computational burden.

 A novel quadratic battery degradation model that can be easily integrated into optimization-based management strategies. This model can be applied to both lead-acid and lithium-ion batteries.

The remainder of this paper is organized as follows: Section II describes the active power dispatch problem considering the long-term management of storage systems. Section III presents the complete power flow model considering reactive power, as well as the magnitude and angle of voltages, and the inclusion of ancillary services. The degradation model is presented in Section IV, where the tuning process of the model's parameters (which depends on the storage system technology) is also shown. In addition, Section IV describes the integration of the degradation model to the formulation of the active power dispatch problem. Section V presents the OPF algorithm based on ADMM and consensus. and Section VI shows the simulation scenario and the case studies, where a co-simulation scheme between Matlab and DIgSILENT is used to validate the performance of the algorithm and obtain numerical results. Finally, conclusions are drawn in Section VII.

## II. ACTIVE POWER DISPATCH CONSIDERING STORAGE SYSTEMS

#### A. PRELIMINARIES

#### 1) MICROGRIDS AS GRAPHS

A microgrid has buses (i.e., physical connection points for generators, loads, and transformers) interconnected by distribution lines. Thus, a microgrid can be modelled by an undirected graph  $\mathcal{G} = (\mathcal{N}, \mathcal{L})$ , where  $\mathcal{N} = 1, \ldots, N$  is the set of nodes (buses) and  $\mathcal{L} \in N \times N$  is the set of arcs (lines), such as, if buses *i* and *j* are linked by a distribution line, then  $(i, j) \in \mathcal{L}$ .

The nodes have some associated variables. For instance, all nodes have a voltage magnitude  $V_i$  (where *i* is the node index) and an angle  $\theta_i$ . Moreover, if a node has a generator, it has active and reactive power generation, denoted by  $P_{G_i}$  and  $Q_{G_i}$ , respectively. Finally, if an electrical load is connected to a node, it has active consumption  $P_{L_i}$  and reactive consumption  $Q_{L_i}$ .

#### 2) POWER FLOW

The power flow (PF) is a feasible solution to the problem of satisfying the power demands of the electrical loads using generators through active and reactive power injections. The power flow in a polar form is defined as

$$P_i(V, \theta) = P_{G_i} - P_{L_i}, \forall i \in \mathcal{N},$$
  
$$Q_i(V, \theta) = Q_{G_i} - Q_{L_i}, \forall i \in \mathcal{N},$$

where  $P_i(V, \theta)$  and  $Q_i(V, \theta)$  are active and reactive injections [26]. If flow is positive the node is providing power,

VOLUME 11, 2023

else it is consuming. These variables are expressed in terms of magnitude and angle of voltage as

$$P_i(V,\theta) = V_i \sum_{j \in \mathcal{N}_i} V_j(G_{ij} \cos \theta_{ij} + B_{ij} \sin \theta_{ij}), \qquad (1)$$

$$Q_i(V,\theta) = V_i \sum_{j \in \mathcal{N}_i} V_j(G_{ij} \sin \theta_{ij} - B_{ij} \cos \theta_{ij}), \qquad (2)$$

for all  $i \in \mathcal{N}$ , and where  $G_{ij}$  and  $B_{ij}$  are elements of the admittance matrix related with conductance and susceptance of line (i, j). The variable  $\theta_{ij}$  represents the difference between angles  $\theta_i$  and  $\theta_j$ , and the notation  $\mathcal{N}_i$  represents the set of nodes connected to the  $i^{th}$  node, including *i*.

The approximations for the energy constraints in a grid presented in (1) and (2) are often combined with an objective cost function establishing the optimal power flow (OPF) in microgrids [27], [28]. Next, we propose two joint optimization problems to deal with technical and economic objectives, providing, in addition, some ancillary services and a strategy for storage management.

#### B. ACTIVE-POWER DISPATCH PROBLEM CONSIDERING BATTERY ENERGY-STORAGE SYSTEMS

Microgrids integrate battery energy storage systems (BESS) to support intermittent generation units, coordinated jointly with conventional generators by operation schemes such as the economic dispatch (ED), active power dispatch, or other variants. BESSs are modeled by accumulation dynamics that require multi-temporal management, adding a significant computational cost depending on the programming horizon.

The active power dispatch with BESSs determines the generation of each unit, including storage systems, to minimize the operating cost. It is mandatory to satisfy the restriction of supplying the aggregate demand and the constraints related to the generation limits. Besides, BESSs require the management of charge/discharge actions to keep the state of charge (SOC) of batteries within operational limits.

Consider a microgrid with a set of buses  $\mathcal{N}$ , containing a set of generators  $\mathcal{N}_G = \{1, 2, \dots, N_G\}$ , and a set of BESSs  $\mathcal{N}_B = \{N_G + 1, N_G, +2, \dots, N_G + N_B\}$ . For the generation unit  $i \in \mathcal{N}_G \cup \mathcal{N}_B$  (battery or generator), the active power profile is given by the vector  $P_{G_i} = [P_{G_i}^1, P_{G_i}^2, \dots, P_{G_i}^K]$ during a multi-time period  $\mathcal{K} = \{1, 2, \dots, K\}$ . In this period, generators and BESSs seek to coordinate the active dispatch and charging/discharging actions to solve

$$\min_{P_{G_i}} \quad \mathcal{F} = \sum_{i \in \mathcal{N}_G \cup \mathcal{N}_B} \mathcal{F}_i(P_{G_i})$$

subject to

$$\sum_{i \in \mathcal{N}_G \cup \mathcal{N}_B} P_{G_i}^k = P_L^k,\tag{3}$$

$$P_{G_{imin}} \le P_{G_i}^k \le P_{G_{imax}}, \ \forall i \in \mathcal{N}_G \cup \mathcal{N}_B, \tag{4}$$

$$S_j^{k+1} = S_j^k + P_{G_j}^k, \quad \forall j \in \mathcal{N}_B,$$
(5)

$$S_{jmin} \le S_j^k \le S_{jmax}, \qquad \forall j \in \mathcal{N}_B,$$
 (6)

31481

for all  $k \in \mathcal{K}$ , where  $\mathcal{F}$  is a cost function that captures the generation and storage cost,  $P_L^k$  is the aggregate power demand at time k,  $P_{Gimin}$ ,  $P_{Gimax}$ ,  $S_{jmin}$ ,  $S_{jmax}$  are the operational limits associated with the generators units and BESSs, and  $S_j^k$  represents the SOC of BESS  $j \in \mathcal{N}_B$  at time k, defining  $S_j^1$  and  $S_j^{K+1}$  as the initial and final states of charge. We have not considered startup/shutdown and ramp rate constraints because our model assumes that both generators and batteries have fast responses [29], [30]. Nevertheless, we recommend considering startup/shutdown and ramp rate constraints for large-scale microgrids whose generators have slow dynamics [31], [32].

The cost function  $\mathcal{F}$  considers generation cost and battery degradation cost by

$$\mathcal{F} = \sum_{i \in \mathcal{N}_G} F_i(P_{G_i}) + \sum_{j \in \mathcal{N}_B} \hat{F}_j(P_{G_j}), \tag{7}$$

where  $F_i(P_{G_i})$  is a quadratic cost function of  $i^{th}$  generator or BESS defined by  $\sum_{k \in \mathcal{K}} (a_i + b_i P_{G_i}^k + c_i (P_{G_i}^k)^2)$  with  $a_i, b_i, c_i$ cost coefficients. The function  $\hat{F}_j(\cdot)$  is the cost of degradation of the  $j^{th}$  BESS described by  $B_{c,j} \sum_{\delta \in \Delta} f_c(C_j^{\delta}) f_d(D_j^{\delta})$  where:  $B_{c,j}$  is the battery cost,  $f_c(C_j^{\delta})$  and  $f_d(D_j^{\delta})$  are capacity-loss functions due to the number of cycles  $C_j^{\delta}$  and DODs  $D_j^{\delta}$ that occurs in the programming horizon, more details are provided by (20) in Section IV. We assume that  $\mathcal{F}$  is convex to guarantee a global minimum in the problem. Moreover, for each BESS  $j \in \mathcal{N}_B$  in (5), a positive value of  $P_{G_j}^k$  indicates that the battery is charging with a power equal to  $P_{G_j}^k$ , whereas a negative value indicates discharge.

#### **III. OPF AND ANCILLARY SERVICES**

Optimal power flow mixes PF equations with an objective function to set an optimization problem to improve the performance of the power system or different technical requirements [33]. We establish an optimization problem based on OPF to adjust the reactive powers and voltage profiles, so that the objectives of reducing losses in distribution lines and maintaining voltages close to 1 p.u. are met. These ancillary services enhance the level of stability and improve the efficiency of the microgrid.

Voltages and power losses are sensitive to active and reactive power injections. These relations are well-established by the PF in (1), (2) and the power losses are given

$$P_{loss} = \sum_{(i,j)\in\mathcal{L}} G_{ij}(V_i^2 + V_j^2 - 2V_iV_j\cos\theta_{ij}).$$
(8)

Notice that voltages and power losses depend on the power flows through the lines and are calculated according to the parameters of the distribution lines (resistance, reactance, conductance, and susceptance). Additionally, voltage magnitude and angle are not absolutely decoupled. These variables are related by quadratic functions that make the classical PF nonlinear, complex, and low-efficient [34]. In order to formulate an optimization problem considering a complete PF model, we use a linear PF with decoupled variables, high accuracy, and computational efficiency.

#### A. LINEARIZED PF

We use a linearized and decoupled PF introduced in [35]. The approximation is based on the fact that the differences between the phase angle do not exceed 20 degrees and voltage magnitudes are 1 p.u. approximately, for most small-scale power systems. In addition, the lengths of the lines are smaller than the ones in large-scale systems (transmission systems), so shunt susceptance or line capacitance can be neglected.

According to the notation of the PF linearization process presented in the Appendix A, the active power flow in (1) can be expressed as

$$P_i(V,\theta) = \sum_{j \in \mathcal{N}_i} G_{ij} V_j - \sum_{j \in \mathcal{N}_i} B_{ij} \theta_j, \qquad (9)$$

where  $\mathcal{N}_i$  is the set of neighbors of the node  $i^{th}$ . In a similar manner, the reactive power flow in (2) can be established as

$$Q_i(V,\theta) = -\sum_{j \in \mathcal{N}_i} B_{ij} V_j - \sum_{j \in \mathcal{N}_i} G_{ij} \theta_j.$$
(10)

The PF model in (9) and (10) is completely linear and decoupled, with the advantage that it does not require an iterative process to find a solution. This relaxation allows the problem to integrate the PF in the constraints of the optimization, mantaining the desirable relationship between voltages and active and reactive power flows. In addition, this model can be used in radial, and meshed distribution systems conserving the accuracy and improving computation efficiency [34].

#### **B. ANCILLARY SERVICES**

Distributed generation can enable ancillary services (AS) such as voltage regulation, congestion relief, selfconsumption, and energy displacement. These ASs help to face several common issues of distribution systems to maintain the integrity, stability, and power quality [4]. For instance, voltage variations on distribution levels can cause stability issues. When demand in a microgrid is too high, voltage profiles may drop below desirable limits, while overproduction of renewable generators produces over-voltages. In the same way, higher consumption leads to higher power losses and peak power demands, increasing electricity energy costs.

Providing AS in a microgrid with distributed energy resources (DERs) is challenging due to uncertainties associated with intermittent distributed generators (DGs), bidirectional flow, and limited reserves. BESSs show great potential to engage AS, stabilizing intermittent generation and increasing flexibility in balancing supply and demand. In addition, power electronic interfaces (inverters and converters) are enhancing their P-Q capabilities to provide AS by reactive power injections [36]. Although power electronic interfaces do not have the robust capacity curves of the synchronous generators, they can generate active and reactive power operating points in a range within a smaller P-Q curve. Voltage regulation can be performed with BESSs and generators using an appropriate reactive power interface. Analogously, power losses can be reduced by reactive power injections, which adjust voltage-node profiles and reduce the total power losses. These services can be provided by an optimization strategy, in which, distribution losses and voltage deviation can be considered into the objective function as

$$\min_{V_i, \mathcal{Q}_{G_i}} \quad H = w_l P_{loss} + w_v \sum_{i \in \mathcal{N}} (1 - V_i)^2 \tag{11}$$

subject to

Linearized PF Equations (9) and (10),  $\forall i \in \mathcal{N}$ ,

$$Q_{G_{imin}} \le Q_{G_i} \le Q_{G_{imax}}, \quad \forall i \in \mathcal{N}_G \cup \mathcal{N}_B, \tag{12}$$

$$V_{imin} \le V_i \le V_{imax}, \quad \forall i \in \mathcal{N},$$
 (13)

$$\theta_{imin} \le \theta_i \le \theta_{imax}, \quad \forall i \in \mathcal{N},$$
(14)

where the objective function H is composed of two terms; the first one refers to the distribution losses (8), and the second term is a summation of penalties for voltage deviations in the nodes. Each form is weighted by parameters  $w_l$  and  $w_v$ . The decision variables are the voltages  $(V_i)$  and reactive powers of the generators and batteries  $(Q_{G_i})$ . The voltages define an optimum operation point to minimize the objective function that can be reached through reactive power variation in generators. The PF equations (9) and (10) define the relationship between reactive power and the magnitude and angle of the voltages at each node. In (9), the active power is not a decision variable since it can be considered a fixed value, which is the solution of the economic dispatch. The constraints (12) - (14) are related to the reactive power restrictions of the generators, BESSs, and the magnitudes and angles of the voltages. These limits can be set up according to each region's electrical standards and regulatory aspects.

From a practical point of view, most of renewable generators and BESSs have no rotational inertia, but some power interfaces enable them to maintain the voltage and frequency stability in isolated microgrids [37]. For example, some inverters can establish active and reactive power flows through power factor. The constraints in the operation point are respresented by a capability curve, which is given by [38]

$$\Phi = \cos^{-1} (PF),$$
  
$$|Q(t)| = P(t) \tan (\Phi),$$
 (15)

where PF is the power factor of the power interface,  $\Phi$  is the power angle, and P(t) and Q(t) are the active and reactive powers at the instant *t*. Notice that the maximum and minimum reactive powers are conditioned to the active power flow and depend on the PF.

#### **IV. BATTERY DEGRADATION MODEL**

Batteries enable the integration of PV systems in microgrids, increasing safety, flexibility, reliability, and quality by performing ancillary services. Energy management models allow performing ancillary services with batteries, like voltage regulation, frequency regulation, peak shaving, and flexibility in balancing supply and demand [4]. However, most of these models consider only basic ancillary services and only a few consider a battery degradation model. Then, current models lead to an incomplete overview of ancillary services enabled with batteries.

Most of the energy management systems of batteries are based on policies if-then, which do not provide an optimal result. Consequently, batteries are charged when generation is high and discharged when generation is low, generating an overuse of BESSs. On the other hand, recent studies consider aging models as part of the energy management model to obtain the scheduling of charge/discharge batteries. Results show that including the effects of degradation reduces the cost of operation (15% approx.). Furthermore, charge/discharge actions are significantly affected, reducing the number of cycles performed per day but maintaining the amount of energy exchanged [39].

A promising research field includes proposing strategies to mitigate the battery degradation while undertaking other issues such as time-consuming, in-line execution, providing ancillary services, and testing different optimization algorithms [22], [40]. However, little attention has been paid to the reliability and accuracy of the battery degradation model, which leads to batteries to work in non-optimal operating points. Moreover, many degradation models are not validated in practice or are only based on battery datasheet curves like the cycle-vs-DOD (depth of discharge) charts, which are obtained under controlled testing conditions instead of real scenarios. Although a few studies validate the battery degradation models, most of them are too complex to be integrated into some optimization problems [41] because they include discrete variables, discontinuities, or highly nonlinear terms.

We propose a new battery degradation model so-called Quadratic Ah-throughput, a quadratic model to estimate battery degradation for lead-acid and lithium-ion batteries. Besides, the model can be easily integrated into the optimization formulation to obtain fast solutions. Remarkably, the Quadratic Ah-throughput model is based on a structure validated in real scenarios with high accuracy on estimation. Furthermore, the proposed model allows the management system to assess the influence of battery aging in performing ancillary services, such as reducing voltage deviation, distribution power losses, and balancing supply and demand. In this case, the battery degradation model is incorporated in the tertiary control level, specifically in the cost function (7) as a penalizing term of aging.

#### A. QUADRATIC AH-THROUGHPUT MODEL

The proposed model is a quadratic version of the Schiffer model, described in [42], that maintains a high estimation accuracy. In numbers, the weighted Ah-throughput model of Schiffer had an error of 6% on assessment whereas other models (e.g. models based on cycles, DOD and energy exchanged) had the error above 60% [21] and [22]. Besides,

given that Schiffer model is highly complex, we extract only the main features and variables to build a reliable scheme.

The quadratic Ah-throughput (QAh) is a differential model to calculate the total degradation  $(Q_t(\bar{k}))$  or the remaining battery capacity up to instant  $\bar{k}$ . The total degradation is calculated in terms of the past instant and the change of degradation  $(Q_t(\bar{k} - 1), \Delta Q(\bar{k}))$  at the instant  $\bar{k}$ . The change of degradation is defined by two losses functions:  $f_c(\cdot)$  and  $f_d(\cdot)$ , so-called cycle-loss function and DOD-loss function, respectively. The QAh model is described by

$$Q_t(\bar{k}) = Q_t(\bar{k} - 1) - \Delta Q(\bar{k}),$$
 (16)

$$\Delta Q(\bar{k}) = f_c(C(\bar{k}))f_d(D(\bar{k})). \tag{17}$$

The cycle-loss function is an exponential function in terms of the number of accumulated cycles  $C(\bar{k})$  up to instant  $\bar{k}$ . Similarly, the DOD-loss function is quadratic on DOD  $D(\bar{k})$ at instant  $\bar{k}$ . Due to the number of cycles and the DOD are slow-changing variables,  $\bar{k}$  is a considerable large period to capture the variation of  $C(\cdot)$  and  $D(\cdot)$ . In this sense,  $\bar{k}$  can be a period of several hours or days (one day is a typical value).

The loss functions are defined by

$$f_c(C(\bar{k})) = \alpha_1 C(\bar{k}) e^{\beta_1 C(k)},\tag{18}$$

$$f_d(D(\bar{k})) = \alpha_2 D(\bar{k})^2 + \beta_2 D(\bar{k}) + \gamma_2,$$
 (19)

where both functions are fitted by coefficients ( $\alpha_1$ ,  $\beta_1$ ) and ( $\alpha_2$ ,  $\beta_2$ ,  $\gamma_2$ ). These coefficients are calculated by an optimization algorithm to maintain the estimation accuracy. The process requires a real value of battery degradation ( $Q_r(t)$ ), provided according to the Schiffer model ([42]) for lead-acid batteries. For lithium-ion batteries, the real degradation value can be provided by experimental data. To observe the validity of the proposed model, the real value is compared to the estimated value of battery degradation calculated by the QAh model in (16) and (17). Both degradation models, Schiffer and (16), are mapped for a range of values of DOD ( $D^m$ ) and number of cycles ( $C^n$ ) conforming a 3D function, whereas the solution of the optimization problem calculates the coefficients ( $\alpha$ ,  $\beta$ ,  $\gamma$ ). The optimization process is described by

$$\min_{\alpha,\beta,\gamma} \hat{e} = \sum_{n=1}^{N} \sum_{m=1}^{M} \left( Q_r(C^n, D^m) - Q_t(C^n, D^m) \right)^2$$
  
Subject to $\alpha, \beta, \gamma \in R$ ,

where the objective function corresponds to the quadratic error  $(\hat{e})$  between real and estimated degradation values. The concavity of the problem eases the search of the optimal value, which can be found with any optimization tool.

The QAh model extracts only the main features of battery degradation. Although some models are described by many variables such as temperature, battery voltage, discharge current, number of cycles, and SOC (state of charge), only a few variables considerably influence the degradation. This simplification can be observed in similar models which use only one or two variables, mainly SOC, DOD, and cycles [21]. In this case, the QAh model uses DOD that captures the minimum

31484

value of SOC and represents the corrosion and degradation effect. Whereas, the number of cycles directly represents the effect of degradation. Moreover, the product of both loss functions (17) allows us to represent the lost capacity for a specific DOD according to the number of accumulated cycles at an instant  $\bar{k}$ . This approach is similar to the Throughput models of [21] but represented in a quadratic form (19).

Other models described by one o two variables are inexact because they are only based on battery datasheet curves. Hence, they are tuned according to test cycles performed by the manufacturer. So, these models do not consider other degradation boosting factors such as voltage levels and discharging currents. Despite of the QAh model is a two-variables model, it is tuned according to the Schiffer model for lead-acid batteries, regarding all degradation factors obtained under real operational scenarios.

The QAh model is integrated into the active power dispatch problem in the degradation cost function  $\hat{F}_j(P_{G_j})$  in (7). The corresponding SOC for each battery profile  $P_{G_j}$  is calculated according to (5). Next, the battery SOC is used for calculating the number of cycles and the DODs occurring during a period K + 1 as shown the Figure 1. These variables are collected in a vector of cycles  $C_j = [C_j^1, C_j^2, \dots, C_j^{\Delta}]$ , and its related vector of DODs  $D_j = [D_j^1, D_j^2, \dots, D_j^{\Delta}]$ , where  $\Delta$  is the vector length. Using the results from (18) and (19),  $\hat{F}_j(\cdot)$  is defined as

$$\hat{F}_j(P_{G_j}) = B_{c,j} \sum_{\delta \in \Delta} f_c(C_j^\delta) f_d(D_j^\delta).$$
<sup>(20)</sup>



FIGURE 1. Representation of cycles and DODs that occurs during a programming horizon.

Then,  $\hat{F}_j(\cdot)$  is the summation of the capacity losses reached during the programming period  $\{1, \ldots, K+1\}$  multiplied by the battery cost  $B_{c,j}$ . It must be highlighted that the cycle is an accumulative variable that must be stored and updated in each optimization process, which is particularly important in a receding-horizon method.

# V. DISTRIBUTED OPERATION BASED ON ADMM AND CONSENSUS DYNAMICS

#### A. PRELIMINARIES

1) ALTERNATING DIRECTIONS METHOD OF MULTIPLIERS

Alternating directions method of multipliers (ADMM) is an optimization technique that splits the global problem into subproblems that are individually solved. In addition, individual solutions interact in a coordinated way to find a solution to the global optimization problem [43]. From a technical perspective, this method can be understood as a mix that takes advantage of the benefits of dual decomposition and the augmented Lagrangian method to improve convergence.

To explain te general method, consider the following optimization problem with a separable objective function f(x)

$$\min_{x_1, x_2, \dots, x_N} f(x) = \sum_{i \in \mathcal{N}} f_i(x_i),$$
  
Subject to 
$$\sum_{i \in \mathcal{N}} Ax_i = c,$$
 (21)

where  $x = [x_1, x_2, ..., x_N]$ , and  $x_i \in \mathbb{R}^n$  are subvectors of  $x, A \in \mathbb{R}^{m \times n}$  and  $c \in \mathbb{R}^m$ . Moreover,  $f_i(\cdot)$  are assumed to be convex, which is necessary to guarantee convergence. The augmented Lagrangian of (21) can be written as

$$\mathcal{L}_{\rho}(x, y) = \sum_{i \in \mathcal{N}} \mathcal{L}_{\rho_i}(x_i, y)$$
  
= 
$$\sum_{i \in \mathcal{N}} \left( f(x_i) + y^{\top} (Ax_i - c) + \frac{\rho}{2} \|Ax_i - c\|_2^2 \right),$$

where  $y \in \mathbb{R}^m$  is the vector of dual variables and  $\rho > 0$  is a penalty parameter. Besides, the last term in the summation is a regularization term that improves the robustness and the convergence rate of the algorithm.

ADMM iteratively solves the N-agents problem in (21) with

$$x_i^{k+1} := \underset{x_i}{\operatorname{argmin}} \mathcal{L}_{\rho_i}(x_i, y^k)$$
(22)

$$y^{k+1} := y^k + \rho(Ax_i^{k+1} - c), \tag{23}$$

where the penalty term  $\rho$  is also the step size. Notice that the ADMM algorithm has one minimization step (22), and one updating step of the Lagrange multipliers (23). Then, the augmented Lagrangian is sequentially minimized in a process where (22) directs the solution towards the optimum in the  $x_1$ -direction, then it repeats the process directing the solution in the  $x_2$ -direction, and so on, until it reaches  $x_N$ -direction. The procedure is repeated until convergence. This process leads the search for a global solution through alternating directions. The convergence of the method is guaranteed due to the convexity of the objective function (7) and the linearity of the constraints (3)-(6). Hence, we can obtain a Lagrangian function whose saddle point is the optimal point over the domain of the primal variables and the multipliers. This point satisfies the Karush-Kuhn-Tucker (KKT) necessary conditions, which are also sufficient since the objective function  $\left(\sum_{i\in\mathcal{N}_G}F_i(P_{G_i})+\sum_{j\in\mathcal{N}_B}\hat{F}_j(P_{G_j})\right)$  is convex.

It is worth noting that the global term in (23) requires a central aggregator to broadcast this information to the other N individual processors. The algorithm is not fully distributed because y is a global variable that is shared by all agents. The next section uses consensus protocols to distribute the global information.

#### 2) ACCELERATED CONSENSUS PROTOCOL

Consensus protocols are iterative algorithms whose goal is reaching the same value in the nodes of the graph (generally, the average of the nodes' initial conditions), using only local information at each updating step. Given an undirected graph G = (V, E) consisting of a set of nodes V = 1, ..., N and a set of edges  $E \subset V \times V$ , with initial conditions  $\lambda^0 \in \mathbb{R}^N$ , the consensus protocol reaches the state  $(1/N)\mathbf{1}^{\top}\lambda^0$ .

This section describes two consensus protocols: the widely used linear consensus protocol and an accelerated version of it [44]. The linear consensus protocol achieves average state following the updating rule

$$\lambda_i^{l+1} = \lambda_i^l + \sum_{j \in \mathcal{N}_i} w_{ij} (\lambda_j^l - \lambda_i^l), \qquad (24)$$

for each node  $i \in V$ , where  $\lambda_i^l$  is the  $i^{th}$  entry of the vector  $\lambda^l$ and  $w_{ij} = w_{ji}$  is the weight associated with the edge  $(i, j) \in E$ . If we let W be the weighting adjacency matrix<sup>1</sup> of G, and defining  $\mathcal{W} \in \mathbb{R}^{N \times N}$  as the matrix whose entries satisfies  $[\mathcal{W}]_{ij} = [W]_{ji}$  if  $i \neq j$ , and  $[\mathcal{W}]_{ii} = 1 - \sum_{j \in \mathcal{N}_i} W_{ij}$ ; then the consensus protocol in (24) can be expressed in the compact form

$$\lambda^{l+1} = \mathcal{W}\lambda^l.$$

The accelerated consensus protocol (ACP) is a two-step algorithm where each node  $i \in V$  updates two variables  $\lambda_i^l, \mu_i^l$  at each iteration *l*. The update rule is given by

$$\mu^{l+1} = \mathcal{W}\lambda^{l},$$
  

$$\lambda^{l+1} = \mu^{l+1} + \frac{l+1}{l+3}(\mu^{l+1} - \mu^{l}),$$
(25)

where  $\mu^l = [\mu^1, \dots, \mu^N]^\top$  and  $\mu^0 = W\lambda^0$ .

The accelerated consensus protocol has a linear convergence rate, which is highly convenient to face multi-temporal problems as scheduling of batteries. More details of the accelerated consensus protocol can be found in [44].

#### B. ADMM AND ACP FOR SOLVING THE ACTIVE-POWER DISPATCH WITH BESSS

Our purpose is to use ADMM and ACP to obtain an optimal solution for the problem in (3)-(7) in a fully distributed way without a central connector node.

First, we define the augmented Lagrangian of (7) as

$$\mathcal{L}_{\rho}(P_{G}, Y, E) = \sum_{i \in \mathcal{N}_{G} \cup \mathcal{N}_{B}} \mathcal{L}_{\rho_{i}}(P_{G_{i}}, Y, E)$$
$$= \sum_{i \in \mathcal{N}_{G} \cup \mathcal{N}_{B}} \left( \mathcal{F}_{i}(P_{G_{i}}) + Y^{\top}E + \frac{\rho}{2}EE^{\top} \right),$$
(26)

where  $\mathcal{F}_i$  is the objective function of the *i*<sup>th</sup> generation unit (generator or BESS), Y is the Lagrange-multipliers vector, and E is a vector of power balance constraints

<sup>1</sup>The entries of *W* are as follows:  $[W]_{ij} = w_{ij}$  if  $(i, j) \in E$ , and  $[W]_{ij} = 0$  otherwise.

(which need global information) given by

$$E = \begin{bmatrix} \sum_{i \in \mathcal{N}_{G}} P_{G_{i}}^{1} + \sum_{j \in \mathcal{N}_{B}} P_{G_{j}}^{1} - P_{L}^{1} \\ \sum_{i \in \mathcal{N}_{G}} P_{G_{i}}^{2} + \sum_{j \in \mathcal{N}_{B}} P_{G_{j}}^{2} - P_{L}^{2} \\ \vdots \\ \sum_{i \in \mathcal{N}_{G}} P_{G_{i}}^{K} + \sum_{j \in \mathcal{N}_{B}} P_{G_{j}}^{K} - P_{L}^{K} \end{bmatrix}.$$
 (27)

Indeed, the elements of E correspond to the power balance in (3) for all K timeslots.

The Lagrangian in (26) can be conveniently expressed to distribute its calculation as

$$\mathcal{L}_{\rho}(P_G, Y, E) = \sum_{i \in \mathcal{N}_G \cup \mathcal{N}_B} \mathcal{L}_{\rho_i}(P_{G_i}, Y_i, E_i)$$
$$= \sum_{i \in \mathcal{N}_G \cup \mathcal{N}_B} \left( \mathcal{F}_i(P_{G_i}) + Y_i^\top E_i + \frac{\rho}{2} E_i E_i^\top \right)$$
(28)

where *Y* and *E* can be calculated and distributed by a central agent. However, the exchange of information with neighboring agents allows the calculation of *Y* and *E* to be distributed by a consensus protocol, generating the correspondingly  $E_i$ ,  $Y_i$  vectors for each agent.

Algorithm 1 describes the fully distributed method based on ADMM and ACP. Iterations are denoted by superscript t and the stopping criteria is reached when the 2-norm of the vector  $E_i^t$  is less than the threshold error  $\epsilon = 1 \times 10^{-3}$ .

# Algorithm 1 ADMM for Active Power Dispatch With BESSs1: $P_{G_i}^{t+1} \leftarrow \underset{P_{G_i}}{\operatorname{argmin}} \mathcal{L}_{\rho_i}(P_{G_i}, Y_i^t, E_i^t), \quad \forall i \in \mathcal{N}_G \cup \mathcal{N}_B$

2:  $\lambda_i \leftarrow \lambda_i^L$ ,  $\forall i \in \mathcal{N}_G \cup \mathcal{N}_B$ , where  $\lambda_i^L$  is the solution of the distributed algorithm (25) at time *L*, with initial conditions  $\lambda_i^0 = \begin{bmatrix} P_{G_1}^{t+1}, \dots, P_{G_{N_G}+N_R}^{t+1} \end{bmatrix}$ ,

3: 
$$E_i^{t+1} \leftarrow N\lambda_i$$
,  $\forall i \in \mathcal{N}_G \cup \mathcal{N}_B$   
4:  $Y_i^{t+1} \leftarrow Y_i^t + \rho E_i^{t+1}$ ,  $\forall i \in \mathcal{N}_G \cup \mathcal{N}_B$ 

where L is the number of iterations at consensus is reached, which is tuned experimentally.

Notice that the major issue in distributing the ADMM is the aggregated load profile  $P_L$  in the vector E in (27), which is a global variable. Nonetheless, ACP generates an individual vector  $E_i$  with local information, making it available for all agents of the microgrid.

The algorithm starts obtaining the optimal power profile for the  $i^{th}$  generator or battery minimizing its corresponding augmented Lagrangian. In case of batteries, the power profile is an optimal charging/discharging actions profile. It is remarkable that in Line 1, each agent (a generator or a battery) proposes an optimal solution to minimize its corresponding local function sequentially and individually.

Line 2 shows how to share global information in a distributed way. First, the consensus protocol's initial conditions are loaded in  $\lambda_i^0$  for each agent and one of the agents must load the aggregated load profile  $P_L$ . For instance, the last agent can load its initial condition as  $\lambda_N^0 \leftarrow P_{G_N}^{t+1} - P_L$ . In any case, the aggregated load profile  $P_L$  is only known by one agent. Conveniently, the proposed selection of the initial conditions leads the consensus to converge to the average of their initial states, which is precisely the vector  $E_i$  divided by the number of agents N.

Note that the accelerated consensus in (25) returns an average state  $\lambda_i$ , which corresponds to the average power balance for all time slots. Therefore, in Line 3, each average state is multiplied by *N* to obtain an estimated value  $E_i$ . Finally, Line 4 shows the updating equation of the Lagrange multipliers for each agent. The optimal values  $P_{G_i}^* := P_{G_i}^{t+1} \forall i \in \mathcal{N}_G \cup \mathcal{N}_B$  are reached according to the stopping criteria, where the Lagrange multipliers tend to be constant.

#### C. ADMM FOR SOLVING THE REACTIVE-POWER DISPATCH ENABLING ANCILLARY SERVICES

Similar to the previous section, the augmented Lagrangian is defined according to the objective function and the corresponding constraints, and finally, the solving algorithm is presented.

First, the H function in (11) is written conveniently to express as a sum of single functions

$$H = \sum_{i \in \mathcal{N}_G \cup \mathcal{N}_B} H_i(V_i)$$
  
=  $w_l \sum_{i \in \mathcal{N}_G \cup \mathcal{N}_B} \left( \frac{1}{2} \sum_{j \in \mathcal{N}_i} G_{ij} (V_i^2 + V_j^2 - 2V_i V_j \cos \theta_{ij}) \right)$   
+ $w_v \sum_{i \in \mathcal{N}_G \cup \mathcal{N}_B} (1 - V_i)^2.$ 

So, the augmented Lagrangian of (11) is defined as

$$\mathcal{L}_{q}(Q_{G}, V, Z, U) = \sum_{i \in \mathcal{N}_{G} \cup \mathcal{N}_{B}} \mathcal{L}_{q_{i}}(Q_{G_{i}}, V_{i}, Z_{i}, U_{i})$$
$$= \sum_{i \in \mathcal{N}_{G} \cup \mathcal{N}_{B}} \left( H_{i} + Z_{i}^{\top} U_{i} + U_{i} U_{i}^{\top} \right),$$
(29)

where  $H_i$ ,  $Z_i$ ,  $U_i$  are the objective function, dual variables and constraint vectors of the  $i^{th}$  agent, respectively. The vector  $U_i$  is given by

$$U_i = \begin{bmatrix} \sum_{j \in \mathcal{N}_i} G_{ij} V_j - \sum_{j \in \mathcal{N}_i} B_{ij} \theta_j - (P_{G_i}^1)^* + P_{L_i} \\ - \sum_{j \in \mathcal{N}_i} B_{ij} V_j - \sum_{j \in \mathcal{N}_i} G_{ij} \theta_j - Q_{G_i} + Q_{L_i} \end{bmatrix}$$

 $\forall i \in \mathcal{N}_G \cup \mathcal{N}_B$ . The first term of  $U_i$  is the constraint of active power flow in (9), which includes optimum in the first instant of the active-power dispatch  $(P_{G_i}^1)^*$  found by the Algorithm 1. The second term is the constraint of reactive power flow in (10). Each constraint in the vector  $U_i$  requires only local information since all constraints of  $U_i$  are calculated over the set of neighbors  $\mathcal{N}_i$ .

Algorithm 2 describes the proposed method to obtain the reactive power dispatch based on the ADMM. It works similarly to Algorithm 1, but it calculates the optimal profile of reactive power for generators and batteries by minimizing the augmented Lagrangian  $\mathcal{L}_q$  in (29). It is worth remarking that Algorithm 2 is completely distributed and does not need consensus.

Algorithm 2 ADMM for Reactive Power Dispatch			
1: for $t = 1$ to $T_{max}$ do			
2: $Q_{G_i}^{t+1} := \underset{Q_{G_i}}{\operatorname{argmin}} \mathcal{L}_{q_i}(Q_{G_i}, Z_i^t, U_i^t)$	$, \forall i \in \mathcal{N}_G \cup \mathcal{N}_B$		
3: $Z_i^{t+1} = Z_i^t + qU_i^t,$	$\forall i \in \mathcal{N}_G \cup \mathcal{N}_B$		
4: end for			

Algorithm 2 starts finding the reactive power profiles for generators and batteries. Then, the Lagrange multipliers are calculated according to constraints  $U_i$  with a q as step size. The optimal reactive power profiles  $Q_{G_i}^* = Q_{G_i}^{T_{max}}, \forall i \in$  $\mathcal{N}_G \cup \mathcal{N}_B$  are reached at the maximum iteration  $T_{max}$  when the Lagrange multipliers tend to be constant.

#### **D. IN-LINE OPERATION APPROACH**

The energy managing strategies of a microgrid must have the ability to respond to disturbances. The robustness feature is motivated due to some inputs to the algorithms are constantly changing, such as the load profile, solar irradiance, and wind speed, among others variables that have an aleatory nature. Although the low-level control hierarchy of a microgrid allows the system to respond to disturbances, it is crucial to deal with disturbances from upper levels to minimize the control effort at lowers levels.

The proposed approach to manage disturbances is based on feedback by measuring disturbances and updating control signals. The in-line operation relies on the ability of the algorithm to process all operations in a maximum time equal to the updating time. Figure 2 shows a scheme of the local monitoring system that allows each agent to measure local variables (power limits, voltages, active and reactive power, and disturbances) to perform the distributed processing. Then, the control signals are updated (active and reactive power setpoints) and the cycle is repeated continuously.



FIGURE 2. Scheme of the in-line operation of the energy-management system in the microgrid.

To avoid the computational complexity, we use a control scheme that splits the programming period into two segments: short-term and long-term periods (see Figure 3). Based on the multi-horizon model predictive control (MPC) scheme in [45], we define a short-term period with  $k_s$  time intervals of width  $\Delta k_s$ . A long-term period with K time intervals is also used with an interval width  $\Delta K$ . Specifically, we propose that the measurements must be updated in the first interval of the short-term period.



FIGURE 3. In-line signals chart.

n

This multi-horizon scheme allows the management system to set a short sampling time  $\Delta k_s$  to deal with disturbances. At the same time, it maintains a large horizon of programming, to preserve a reasonable computational cost. Notice that this combination enables a short sampling time with a low computational complexity because the long-term period uses fewer decision variables. It should be mentioned that this strategy reduces the computational complexity in return for an increase in the number of estimations. Although the horizon is divided by reducing the number of variables, the estimations must also be increased to the same extent. Then, the number of signal estimators must forecast the measured signals every short sampling time to obtain the estimated signals to run the algorithm along the programming horizons.

Figure 3 illustrates the combination of short and large periods. Specifically, we use a long-term period of 24 hours with time intervals of 1 hour, typically used in dispatch problems. Besides, we use a short-term period of 1 hour of 5-minutes time intervals.

The concept of a two-period programming horizon is implemented to use the advantages of the short-term and long-term periods in the formulation of a new active dispatch problem as

$$\min_{P_{G_i},\bar{P}_{G_i}} \sum_{i\in\mathcal{N}_G\cup\mathcal{N}_B} \left(\mathcal{F}_i(P_{G_i}) + \bar{\mathcal{F}}_i(\bar{P}_{G_i})\right).$$
(30)

The objective function in (30) gathers the generation costs for the long-term period  $(\mathcal{F}_i)$  and short-term period  $(\mathcal{F}_i)$ . The active power profile for the short-term period is given by  $\bar{P}_{G_i} = [\bar{P}_{G_i}^1, \bar{P}_{G_i}^2, \dots, \bar{P}_{G_i}^{K_s}]$ , during a short-time interval  $\mathcal{K}_s = \{1, 2, \dots, K_s\}$  with time intervals equal to  $\Delta k_s$ . Similarly to (7), the cost function  $(\bar{\mathcal{F}})$  is a cost function that captures the generation and storage cost during the short-time period.

In adittion to long-term constraints (3)-(6), the constraints related to the short-term period are

$$\sum_{i \in \mathcal{N}_G \cup \mathcal{N}_B} \bar{P}_{G_i}^{k_s} = \bar{P}_L^{k_s},\tag{31}$$

$$P_{G_{imin}} \leq \bar{P}_{G_i}^{k_s} \leq P_{G_{imax}}, \ \forall i \in \mathcal{N}_G \cup \mathcal{N}_B,$$
(32)

$$S_j^{\kappa_s+1} = S_j^{\kappa_s} + P_{G_j}^{\kappa_s}, \qquad \forall j \in \mathcal{N}_B,$$
(33)

$$\bar{S}_{jmin} \le \bar{S}_j^{\kappa_s} \le \bar{S}_{jmax}, \qquad \forall j \in \mathcal{N}_B,$$
 (34)

 $\forall k_s \in \mathcal{K}_s$ . Similarly to constraints for the long-term period, equations in (31)-(34) represent the power balance, power limits, the SOC dynamics, and SOC limits, for the short-term.

#### **VI. SIMULATIONS AND RESULTS**

To show the implementation of the proposed method, first we present the presented the fitting of QAh-Throughput model for lead-acid and lithium batteries to be incorporated in the optimization model. Secondly, the case study to test the optimization strategy is described with all technical details about the microgrid. Next, the results of the active dispatch are presented, and a comparison of the distributed methods with a centralized one is performed in terms of the algorithm velocity and the optimal value. Furthermore, an analysis of the influence of the degradation model is presented considering three scenarios: i) without a degradation model; ii) considering only battery costs; and iii) with degradation model. The interaction of the lead-acid and lithium technologies and the lifetime extension achieved with the degradation model for both technologies is remarkable. Next, the ancillary services by the injection of reactive power are included with a quantitative comparison of the power losses and voltage deviation. Finally, disturbances management by the in-line operation is presented, whose benefits are quantified in terms of the reduction of computational effort and the error between demand and generation.

#### A. TUNNING OF DEGRADATION MODELS FOR BESS

To incorporate the degradation models in the planning of charge/discharge actions for BESSs, we have tailored the proposed model according to experimental data in [46] and the empirical weighted Ah-Throughput model in [21]. Specifically, the Ah-Throughput model can be fitted for lead-acid and lithium-ion batteries from curves of capacity vs. cycles with other DODs, aging experiments, or manufacturer information. Besides, the proposed solution method can get a fast solution because it is convex on DOD.

#### 1) LEAD-ACID BESS

We tune the model according to the weighted model [21], which is accurate in estimating the capacity loss of lead-acid batteries. However, it is not suitable for real-time optimization due to its complexity.

In order to obtain the coefficients of the QAh-Throughput model in (16) and (17), a fitting process is carried out regarding the data from the experimental Ah-Throughput model (See Figure 4). The obtained coefficients are  $\alpha_1 = 0.4585$ ,  $\beta_1 = 3.321$ ,  $\alpha_2 = 3.529$ ,  $\beta_2 = -0.5762$ ,  $\gamma_2 = 0.0283$ .



**FIGURE 4.** Comparison of the weighted Ah-Throughput model and QAh-Throughput model.

The top of the Figure 4 shows the typical DOD vs cycle curve, where the error to predict the cycles for a specific value of DOD is about 7%. The bottom of the Figure 4 shows the remaining capacity for different values of DOD. The absolute error of the QAh-Throughput model is  $\pm 0.04$ , which corresponds to a percentage error of 1.1%.

#### 2) LITHIUM-ION BESS

The fitting process of the QAh-Throughput model for lithium-ion batteries is carried out with real data from [46]. The model fitting is divided into zone a from 0 to 600 cycles and zone b from 601 cycles to 10000. Zones a and b show different behaviors that a unique QAh-Throughput model cannot represent. Zone a shows a nonlinear and significant decrease in capacity for the first cycles. In contrast, zone b describes a linear reduction of capacity with a low rate (see actual data in Figure 5).

#### TABLE 1. QAh-Throughput coefficients for Lithium Batteries.

Zone	$\alpha_1$	$\beta_1$	$\alpha_2$	$\beta_2$	$\gamma_2$
a	-0.0767e-4	-75.7e-4	-62.1	89.6	-0.798
b	0.121e-4	-2.5e-4	0.0524	0.1	0.025

Table 1 shows the fitting coefficients of the QAh-Throughtput model according to the experimental data of lithium-ion batteries for zone *a* and *b*. The comparison of the fitting and experimental data is shown in Figure 5. The model represents the capacity behavior for both *a* and *b* zones with an absolute error of  $\pm 0.0087$ , representing an 1.6%.

#### B. CASE STUDY

The microgrid simulated is based on the test feeder IEEE 13 bus system with voltage level of 4.16 kV (see Figure 6).



FIGURE 5. Comparison of actual data from [46] and QAh-Throughput model for a lithium-ion battery.

In order to analyze the interaction between renewable energy and storage resources in an isolated mode, we include a diesel generator ( $G_1$ ), a photovoltaic generator ( $G_2$ ), a wind turbine ( $G_3$ ), a lead-acid battery system ( $B_1$ ), and a lithium-ion battery system ( $B_2$ ). The diesel unit is the slack generator that provides the voltage and frequency references. It is worth mentioning that the location and size of generation units are established according [47], to minimize generation costs, power losses, and voltage deviations.



FIGURE 6. Adapted IEEE 13 nodes feeder for the study case.

The electrical lines between nodes are solid, while the communications channels between agents are dashed lines. Table 2 details the capacities and costs for generators and storage systems. The maximum limits of reactive power are calculated according to the generation unit's power factor in (15), according to the active power available at a specific instant.

Figure 7 shows the aggregated load profiles of the active and reactive power, which correspond to actual data of one-week power consumption of a university campus in the Southwest Colombia. The individual load profiles for each node are obtained by dividing the aggregated profiles according to the nominal capacity of each load in Table 3. The solar irradiance and wind speed profiles are shown in Figure 8, which are 7-days real measurements obtained in

#### TABLE 2. Characteristics of the generators and the storage systems.

Generations	Active power	Reactive power	Cost coefficients
units	limits $[kW]$	limits $[kVAr]$	$(c_i; b_i; a_i)$
$G_1$	[200, 2000]	[0, 2000]	0.01; 1.50; 0.5
$G_2$	[0, 1200]	[0, 968]	0.00; 0.50; 0.5
$G_3$	[0, 1000]	[0, 484]	0.00; 0.80; 0.5
$B_1$	[0, 800]	[0, 387]	0.00; 0.25; 0.0
$B_2$	[0, 1200]	[0, 581]	0.00; 0.35; 0.0



FIGURE 7. Aggregated load profiles for the microgrid.



FIGURE 8. Renewable resources profiles for PV and wind generators.

TABLE 3. Nominal capacity of loads.

Node	$\mathbf{kW}$	$\mathbf{kVAr}$
634	400	290
645	170	125
646	230	132
652	128	86
671	1225	717
675	843	462
692	170	151
611	170	80
632	100	58

June of 2021 with a meteorological station located in Pasto - Colombia.

#### C. ACTIVE DISPATCH WITH BESS

Figure 9 shows the active power dispatch obtained with ADMM-ACP (ADMM with accelerated consensus protocol). Clearly, the cheap solar and wind generators ( $G_1$  and  $G_3$ ) are dispatched to their maximum capacity, while the batteries are charged and preserved to provide cheap energy in the rush hours. The expensive diesel generator ( $G_1$ ) completes the rest of the power to supply the aggregate demand in the system. Besides, all active power profiles are within the limits and the power balance is accomplished properly. The performance of the ADMM-ACP is compared with a centralized method (sequential quadratic programming - SQP), the ADMM with aggregator, and the ADMM with consensus (ADMM-C),



FIGURE 9. Active dispatch profiles with BESSs.

 TABLE 4. Performance of the methods.

Method	Cost	Power balance error	Iter.	Time
SQP	1.7396e6	2.63e-12	-	19.37s
ADMM	1.7396e6	9.67e-04	300	67.65s
ADMM-C	1.7406e6	9.99e-04	88	42.67s
ADMM-ACP	1.7396e6	9.10e-04	88	34.68s

without substantial differences between power profiles as it is shown in Table 4. This fact is particularly important since the centralized algorithm relies on complete information to reach the result, while the distributed algorithms uses only local information to achieve the same optimal value. Figure 6 depicts in dotted lines the links of communication between neighbors to perform the distributed techniques.

To guarantee the same effort in the optimization process, we define as stopping criteria the error of the power balance lower or equal to  $1 \times 10^{-3}$  for the distributed methods. Table 4 also shows that the optimal cost values are the same for all methods with a slight difference for ADMM-C. However, the centralized process is less time-consuming than the distributed ones since ADMM updates variables one by one, whereas the centralized method updates all variables simultaneously.

On the other hand, there are essential differences between the convergence time of distributed methods. The ADMM-C and the ADMM-ACP are faster than the classic ADMM with the aggregator. Besides, notice that the ADMM-ACP and the ADMM-C need the same iterations to converge, but the first one uses less time. This difference occurs because ADMM-ACP only needs 20k iterations to reach the consensus, while ADMM-C needs three times more iterations (60k). This reduction of the consensus iterations could be particularly significant in an actual implementation, where the communication delays play an essential role.

## D. EFFECTS OF THE DEGRADATION MODEL IN THE SCHEDULING OF BATTERIES

Three test scenarios are defined to compare the degradationmodel effect on the optimization process. Scenario 1 is the

31490

base case and it considers the batteries as free-of-charge units, i.e., the costs of energy discharged by the batteries are not considered in the optimization function. Then, both units are treated as equals in the optimization process. Scenario 2 regards a price by energy discharged of the batteries with the costs presented in Table 2. In economic terms, the batteries are treated as generators, and their energy costs are considered in the optimization function. Finally, Scenario 3 regards only degradation as a penalization term in the optimization function. The batteries' energy costs are not considered.



FIGURE 10. Loss of capacity for batteries in each scenario.

Figure 10 shows the evolution of batteries' capacity for the three scenarios. From the overall view, the differences between loss-of-capacity curves of lithium-ion and leadacid battery technologies are evident. As expected, the loss of capacity of the lithium-ion battery decreases slowly in comparison with lead-acid technology. In contrast, lead-acid battery capacity decays rapidly and must be replaced several times for Scenario 1 during the time horizon.

Table 5 shows the weeks to end-of-life (EOL) for the three scenarios for each technology. Furthermore, taking as reference the end-of-life of the lithium-ion battery, the table shows the number of replacements of lead-acid batteries per scenario.

TABLE 5. Comparison of duration of batteries.

Scenario	EOL of lithium-ion bat.	EOL of lead-acid bat.	# of replacements of lead-acid bat.
1	630 wks	96 wks	6
2	506 wks	144 wks	3
3	873 wks	248 wks	3

Clearly, the optimization results in Scenario 3 show that the degradation model significantly expands the batteries' lifespan. According to Table 5, the EOL of the lithium-ion battery increases a 38% and a 72% concerning scenarios 1 and 2, respectively. Likewise, the EOL of the lead-acid battery increases a 158% and a 72% in comparison to scenarios 1 and 2. Moreover, the number of replacements of the lead-acid battery required in Scenario 3 are reduced to the half compared with the base case (Scenario 1). This important increase

in the lifespan of the batteries can be explained by the charging/discharging pattern of the batteries.



FIGURE 11. Number of cycles and DOD for each scenario.

Figure 11 shows, in Scenario 1, that both batteries are discharged to similar DOD levels without considering a difference in technologies. This discharging levesl lead to rapid degradation of the lead-acid battery compared to the lithium-ion technology as it is presented in Figure 10. In contrast, Scenario 2 reduces the DOD of lead-acid technology as a result of the economic power dispatch, which expands the lifespan of both battery technologies in comparison with the base case. However, a significant increment in the batteries' life is obtained regarding the model degradation in Scenario 3. To reach this increase, the degradation model reduces the DODs of both batteries and increases the number of cycles, as it is shown in bar graph of Scenario 3 in Figure 11. Furthermore, at the same time, the load's energy requirements are fulfilled.



FIGURE 12. Changes of the SOC of batteries in Scenario 3.

Besides, it is worth highlighting that the degradation model adjusts the DOD according to the state of degradation of the battery along the time, as it is shown in Figure 12. At the start of the battery life (in the first week) lead-acid battery has a deeper degradation rate than lithium-ion technology. As a result, the lead-acid has a higher DOD than the lithium-ion battery in the first weeks. On the contrary, at the EOL of batteries (at weeks 750 and 240) the DOD is higher for lithium-ion batteries than lead-acid ones. This effect occurs because the lead-acid battery has a higher rate of degradation than lithium-ion technology at EOL.

#### E. REACTIVE DISPATCH AND ANCILLARY SERVICES

Algorithms 1 and 2 perform jointly the optimal power flow of the IEEE 13-nodes microgrid in a distributed fashion, guaranteeing that the main variables are within the proper ranges. Figure 13 shows the reactive power of each generation unit with their respective power limits in dashed lines. Notice that reactive power limits of each generator are calculated according the active power available by (15). Specifically, the power factor of each power interface defines the ability to absorb or generate reactive power. This feature is known as the capability curve. Moreover, the last graph of Figure 13 shows the comparison of total reactive generation and the aggregated reactive demand, whose balance is fulfilled.



FIGURE 13. Reactive dispatch.



FIGURE 14. Comparison of voltage deviation.

In this case, the reactive dispatch enables the ancillary services of reducing power losses and voltage deviation, by using the optimization function (11). Figure 14 shows the influence of reactive dispatch in reducing the voltage deviation (VD). The continuous line shows the VD using only the slack generator ( $G_1$ ) to fulfill the reactive power demand. Whereas the dashed line shows the VD when the

other generation units are enabled to adjust the reactive point operation. Notice that the improving of VD is remarkable. For example, the VD in nodes 611, 652, 671 and 680 are 47.9%, 47.1%, 43.1%, respectively, and a 26% in average for all nodes.

On the other hand, Figure 15 shows that power losses are also reduced by reactive power dispatch. In numbers, the power losses are reduced 22.9% by the reactive injections of renewable generators and batteries, which are limited by their capability curve.



FIGURE 15. Comparison of power losses.

#### F. IN-LINE OPERATION

The proposed in-line scheme is based on a receding horizon strategy capable to update the active and reactive power dispatch to response to disturbances. The programming horizon is split in two programming periods: the long-term is 24 hours with hourly slots, and the short-term is 5 minutes with one slot. Users could tune these features according to the available computational capacity and the size of the microgrid. The in-line scheme is suitable to solve the active dispatch with batteries because the long-term period allows the system to anticipate the battery scheduling. Similarly, a long-term period enables batteries to displace energy from valley hours to rush hours and eases the measurement of the batteries' degradation. At the same time, the short-term period allows the management to response to disturbances by changes in the load and resource profiles.

To test the in-line scheme, we define a load disturbance adding noise to the load profile (Figure 16). The noise is obtained from a normal distribution with variance equal



FIGURE 16. Variation of the active power demand and estimation using long-term and short-term horizons.

to 10% of the maximum value of the load profile. Similarly, the disturbances of wind speed and irradiance are obtained following the same criterium. Notice that Figure 16 shows two estimated profiles: hourly and 5-minutes curves, which are estimations of the actual power demand curve. The system can use sophisticated forecasting tools to obtain more exact predictions [48]; but in any case, both curves are used to build the two-level programming horizon as it is shown in the bottom of the Figure 3.

In-line operation requires additional steps to solve the OPF problem over a receding horizon. Firstly, power demand, wind speed, and irradiation profiles are loaded to make the corresponding estimations according to the programming horizon. Second, the first instants of the profiles are updated by real-time measured data in the short-term and long-term periods. Next, Algorithm 1 solves the active power dispatch problem in (3)-(7) obtaining the generation units' power profiles for the programming horizon. Then, Algorithm 2 solves the reactive power dispatch problem in (11)-(14). At this point, the OPF is solved, and the first setpoints of the active and reactive power profiles are sent to secondary and primary controllers to be used as references. Finally, the programming horizon is displaced to the next instant, the SOCs of batteries are updated, and the process starts again in the first step. In summary, Algorithms 1 and 2 run every short-term period, the active and reactive profiles are updated according the programming horizon, batteries' SOCs are updated, the programming horizon is displaced, and the cycle is repeated.



FIGURE 17. Active power dispatch with in-line scheme.

Figure 17 shows the results with in-line scheme. Notice that all generation units modify their dispatches to satisfy the new power demand. The dotted lines represent the variation of the dispatch profiles during the running of the in-line scheme. Variations are proportional to the noise added to the power demand curve, wind speed profile, and irradiation profile. Similarly, this variation causes  $G_2$  not to be dispatched to the maximum power because of the estimation error.

Figure 18 shows the power balance error between generation and power demand. The graph in the top shows the error for the day-ahead dispatch, and the one in the bottom



FIGURE 18. Error between active power demand and generation.

shows the error obtained with the in-line scheme. The error is reduced because the in-line method corrects the generation units' dispatch, although it does not disappear entirely because the in-line scheme takes a while to get the results. Specifically, graph in the bottom shows the error while the in-line algorithm is processing; during this short time, the power balance error still exists. In numbers, the average errors without and with the in-line scheme are  $\pm 83.05 \, kW$  and  $\pm 0.346 \, kW$ , respectively. In any case, these errors are zeroed by the secondary and primary controllers of the microgrid hierarchy, but the control effort is significantly minor with the in-line scheme.

Another worth-mentioning benefit is the computational complexity reduction. In this case, a conventional receding horizon strategy (as MPC) uses a whole programming horizon of 24 hours with time slots of five minutes, i.e., 288 variables per generation unit. Under the same conditions, the in-line scheme requires 24 variables for the long-term period and one additional variable for the short-term period per generation unit. Running both strategies in a computer with a processor Intel(R) Core(TM) i5-10400 CPU @ 2.90GHz, we obtain that the MPC takes 52.53 seconds per iteration, whereas the in-line scheme takes 4.72 seconds.

#### **VII. CONCLUSION**

A distributed algorithm based on ADMM and consensus is presented to obtain an OPF of microgrids with distributed resources. The algorithm provides ancillary services with batteries and renewable sources. The reduction of voltage deviation and power losses by reactive dispatch are tested by simulation. Finally, the algorithm incorporates degradation models for two battery technologies to extend their lifetime and an in-line scheme to deal with inputs disturbances.

The ADMM-ACP is a strategy to solve the OPF of a microgrid with a performance comparable to conventional ADMM but without an aggregator to share global information. ADMM-ACP is a fully distributed optimization scheme that relies on local information converging to a global optimum. These optimization schemes are suitable in scenarios with multiple agents where it is valuable to limit the sharing

VOLUME 11, 2023

of information, manage the bandwidth communication network of a microgrid, reduce the centralized processing, and perform plug-and-play features, among others.

On the other hand, we present a degradation model that, using experimental data or datasheets, can be adapted to both lead-acid and lithium-ion technologies with a short estimation error. Battery capacity loss is a feature sensitive to DOD and charge/discharge actions and knowing this condition could be significant to preserve the battery life by proper scheduling.

Finally, the in-line operation reduces the computation complexity of batteries' active-power-dispatch problem. At the same time, this time division allows the algorithm to deal with disturbances in a short-time horizon and to schedule batteries for the long term by a multi-horizon strategy.

#### APPENDIX A PF LINEARIZATION

Considering some approximations such as  $V_i, V_j \approx 1$ ,  $\sin(\theta) \approx \theta, \cos(\theta) \approx 1$ . The Equation (1) can be rewritten as

$$P_i(V,\theta) = \sum_{j \in \mathcal{N}_i} G_{ij} V_i V_j \cos \theta_{ij} + \sum_{j \in \mathcal{N}_i} B_{ij} V_i V_j \sin \theta_{ij}.$$
 (35)

Assuming the angle differences between neighboring buses are small, the approximations in Equation (35) are

$$P_i(V,\theta) = \sum_{j \in \mathcal{N}_i} G_{ij} V_i V_j + \sum_{j \in \mathcal{N}_i} B_{ij} V_i V_j \theta_{ij}.$$
 (36)

Now, we define an admittance matrix  $Y_{ij} = G_{ij} + jB_{ij}$  with the following structure

$$Y_{ij} = \begin{cases} -y_{ij} & j \neq i\\ y_{ii} + \sum_{j \in \mathcal{N}_i} y_{ij} & j = i, \end{cases}$$
(37)

where  $y_{ij} = g_{ij} + jb_{ij}$  is the admittance of line (i, j), and  $y_{ii} = g_{ii} + jb_{ii}$  is the shunt admittance at bus  $i^{th}$ .

Then, using Equation (37) and expressing V as  $1.0 + \Delta V$ , the first term of Equation (36) becomes

$$\approx \sum_{j \in \mathcal{N}_i, j \neq i} g_{ij} V_i (V_i - V_j) + g_{ii} V_i^2,$$
  

$$\approx \sum_{j \in \mathcal{N}_i, j \neq i} g_{ij} (1 + \Delta V_i) (\Delta V_i - \Delta V_j) + g_{ii} (1 + 2\Delta V_i),$$
  

$$\approx \sum_{j \in \mathcal{N}_i, j \neq i} g_{ij} (V_i - V_j) + g_{ii} V_i.$$
(38)

Note that  $\Delta V_i$  is smaller than  $V_i$ , so  $\Delta V_i^2$  and  $\Delta V_i \Delta V_j$  can be neglected in the approximation process. Using the admittance matrix, Equation (38) is compacted as

$$\sum_{j \in \mathcal{N}_i, j \neq i} g_{ij}(V_i - V_j) + g_{ii}V_i = \sum_{j \in \mathcal{N}_i} G_{ij}V_j.$$
(39)

31493

Following a similar process, we can approximate the second term of Equation (36) as

$$\approx -\sum_{j \in \mathcal{N}_{i}, j \neq i} b_{ij} V_{i} (V_{i} - V_{j}) \theta_{ij} + b_{ii} V_{i}^{2} \theta_{ii}$$
$$\approx -\sum_{j \in \mathcal{N}_{i}, j \neq i} b_{ij} (\theta_{i} - \theta_{j})$$
$$-\sum_{j \in \mathcal{N}_{i}, j \neq i} b_{ij} (\theta_{i} - \theta_{j}) = -\sum_{j \in \mathcal{N}_{i}} B_{ij} \theta_{j}$$
(40)

To accomplish a successfully variable decoupling and linearization of power flow, we prioresse the influence of angles over voltages in Equation (40). So, using the approximations in (39) and (40) the active power flow in (35) is expressed in a linear fashion as

$$P_i(V,\theta) = \sum_{j \in \mathcal{N}_i} G_{ij} V_j - \sum_{j \in i} B_{ij} \theta_j.$$

In a similar way, the reactive power flow in Equation (2) can be established as

$$Q_i(V,\theta) = -\sum_{j \in \mathcal{N}_i} B_{ij} V_j - \sum_{j \in i} G_{ij} \theta_j.$$

#### REFERENCES

- Y. Han, H. Li, P. Shen, E. A. A. Coelho, and J. M. Guerrero, "Review of active and reactive power sharing strategies in hierarchical controlled microgrids," *IEEE Trans. Power Electron.*, vol. 32, no. 3, pp. 2427–2451, Mar. 2017.
- [2] E. Espina, J. Llanos, C. Burgos-Mellado, R. Cárdenas-Dobson, M. Martínez-Gómez, and D. Sáez, "Distributed control strategies for microgrids: An overview," *IEEE Access*, vol. 8, pp. 193412–193448, 2020.
- [3] S. Moayedi and A. Davoudi, "Distributed tertiary control of DC microgrid clusters," *IEEE Trans. Power Electron.*, vol. 31, no. 2, pp. 1717–1733, Feb. 2016.
- [4] L. Maeyaert, L. Vandevelde, and T. Döring, "Battery storage for ancillary services in smart distribution grids," *J. Energy Storage*, vol. 30, Aug. 2020, Art. no. 101524.
- [5] A. Bidram, F. L. Lewis, and A. Davoudi, "Distributed control systems for small-scale power networks: Using multiagent cooperative control theory," *IEEE Control Syst. Mag.*, vol. 34, no. 6, pp. 56–77, Dec. 2014.
- [6] R. Zhang and B. Hredzak, "Distributed dynamic clustering algorithm for formation of heterogeneous virtual power plants based on power requirements," *IEEE Trans. Smart Grid*, vol. 12, no. 1, pp. 192–204, Jan. 2021.
- [7] J. Peng, B. Fan, and W. Liu, "Voltage-based distributed optimal control for generation cost minimization and bounded bus voltage regulation in DC microgrids," *IEEE Trans. Smart Grid*, vol. 12, no. 1, pp. 106–116, Jan. 2021.
- [8] A. Vasilakis, I. Zafeiratou, D. T. Lagos, and N. D. Hatziargyriou, "The evolution of research in microgrids control," *IEEE Open Access J. Power Energy*, vol. 7, pp. 331–343, 2020.
- [9] T. Wu, C. Zhao, and Y.-J.-A. Zhang, "Distributed AC-DC optimal power dispatch of VSC-based energy routers in smart microgrids," *IEEE Trans. Power Syst.*, vol. 36, no. 5, pp. 4457–4470, Sep. 2021.
- [10] K. Wu, Q. Li, Z. Chen, J. Lin, Y. Yi, and M. Chen, "Distributed optimization method with weighted gradients for economic dispatch problem of multi-microgrid systems," *Energy*, vol. 222, May 2021, Art. no. 119898.
- [11] G. Binetti, A. Davoudi, F. L. Lewis, D. Naso, and B. Turchiano, "Distributed consensus-based economic dispatch with transmission losses," *IEEE Trans. Power Syst.*, vol. 29, no. 4, pp. 1711–1720, Jul. 2014.
- [12] H. Pourbabak, J. Luo, T. Chen, and W. Su, "A novel consensus-based distributed algorithm for economic dispatch based on local estimation of power mismatch," *IEEE Trans. Smart Grid*, vol. 9, no. 6, pp. 5930–5942, Nov. 2018.

- [13] H. Xia, Q. Li, R. Xu, T. Chen, J. Wang, M. A. S. Hassan, and M. Chen, "Distributed control method for economic dispatch in islanded microgrids with renewable energy sources," *IEEE Access*, vol. 6, pp. 21802–21811, 2018.
- [14] J. Barco-Jiménez, G. Obando, A. Pantoja, E. Caicedo, and J. Aguado, "Decentralized operation of an isolated microgrid with storage systems using multipliers with alternating directions," *Renew. Energy Power Qual. J.*, vol. 20, pp. 330–335, Sep. 2022.
- [15] C. Bai, Q. Li, W. Zhou, B. Li, and L. Zhang, "Fast distributed gradient descent method for economic dispatch of microgrids via upper bounds of second derivatives," *Energy Rep.*, vol. 8, pp. 1051–1060, Nov. 2022.
- [16] L. Li, M. Dong, D. Song, J. Yang, and Q. Wang, "Distributed and real-time economic dispatch strategy for an islanded microgrid with fair participation of thermostatically controlled loads," *Energy*, vol. 261, Dec. 2022, Art. no. 125294.
- [17] Y. He, W. Wang, and X. Wu, "Multi-agent based fully distributed economic dispatch in microgrid using exact diffusion strategy," *IEEE Access*, vol. 8, pp. 7020–7031, 2020.
- [18] F. Mehmood, B. Khan, S. M. Ali, and J. A. Rossiter, "Distributed MPC for economic dispatch and intermittence control of renewable based autonomous microgrid," *Electr. Power Syst. Res.*, vol. 195, Jun. 2021, Art. no. 107131.
- [19] S. Negri, F. Giani, A. M. Pavan, A. Mellit, and E. Tironi, "MPC-based control for a stand-alone LVDC microgrid for rural electrification," *Sustain. Energy, Grids Netw.*, vol. 32, Dec. 2022, Art. no. 100777.
- [20] E. Lobato, L. Sigrist, A. Ortega, A. González, and J. M. Fernández, "Battery energy storage integration in wind farms: Economic viability in the Spanish market," *Sustain. Energy, Grids Netw.*, vol. 32, Dec. 2022, Art. no. 100854.
- [21] R. Dufo-López, J. M. Lujano-Rojas, and J. L. Bernal-Agustín, "Comparison of different lead–acid battery lifetime prediction models for use in simulation of stand-alone photovoltaic systems," *Appl. Energy*, vol. 115, pp. 242–253, Feb. 2014.
- [22] S. Wang, D. Guo, X. Han, L. Lu, K. Sun, W. Li, D. U. Sauer, and M. Ouyang, "Impact of battery degradation models on energy management of a grid-connected DC microgrid," *Energy*, vol. 207, Sep. 2020, Art. no. 118228.
- [23] C. Bordin, H. O. Anuta, A. Crossland, I. L. Gutierrez, C. J. Dent, and D. Vigo, "A linear programming approach for battery degradation analysis and optimization in offgrid power systems with solar energy integration," *Renew. Energy*, vol. 101, pp. 417–430, Feb. 2017.
- [24] J. Nocedal and S. J. Wright, *Numerical Optimization*. New York, NY, USA: Springer, 2006.
- [25] P. Spellucci, "A new technique for inconsistent QP problems in the SQP method," *Math. Methods Oper. Res.*, vol. 47, no. 3, pp. 355–400, Oct. 1998.
- [26] Z. Yan and Y. Xu, "Real-time optimal power flow: A Lagrangian based deep reinforcement learning approach," *IEEE Trans. Power Syst.*, vol. 35, no. 4, pp. 3270–3273, Jul. 2020.
- [27] Y. Jia, Z. Y. Dong, C. Sun, and K. Meng, "Cooperation-based distributed economic MPC for economic load dispatch and load frequency control of interconnected power systems," *IEEE Trans. Power Syst.*, vol. 34, no. 5, pp. 3964–3966, Sep. 2019.
- [28] N. Meyer-Huebner, M. Suriyah, and T. Leibfried, "Distributed optimal power flow in hybrid AC–DC grids," *IEEE Trans. Power Syst.*, vol. 34, no. 4, pp. 2937–2946, Jul. 2019.
- [29] D. Cormode, A. D. Cronin, W. Richardson, A. T. Lorenzo, A. E. Brooks, and D. N. DellaGiustina, "Comparing ramp rates from large and small PV systems, and selection of batteries for ramp rate control," in *Proc. IEEE* 39th Photovolt. Spec. Conf. (PVSC), Jun. 2013, pp. 1805–1810.
- [30] V. Gevorgian and D. Corbus, "Ramping performance analysis of the Kahuku wind-energy battery storage system," Denver West Parkway, Golden, CO, USA, Tech. Rep. NREL/MP-5D00-59003, 2013.
- [31] H. Qiu, W. Gu, W. Sheng, L. Wang, Q. Sun, and Z. Wu, "Resilienceoriented multistage scheduling for power grids considering nonanticipativity under tropical cyclones," *IEEE Trans. Power Syst.*, early access, Aug. 31, 2022, doi: 10.1109/TPWRS.2022.3203066.
- [32] J. Hurtt and K. Baker, "Sensitivity analysis of photovoltaic system design parameters to passively mitigate ramp rates," *IEEE J. Photovolt.*, vol. 11, no. 2, pp. 545–551, Mar. 2021.
- [33] S. Frank and S. Rebennack, "An introduction to optimal power flow: Theory, formulation, and examples," *IIE Trans.*, vol. 48, no. 12, pp. 1172–1197, Aug. 2016.

- [34] Y. C. C. Wong, C. S. Lim, H. H. Goh, A. Cruden, M. D. Rotaru, and X. Kong, "An optimal secondary multi-bus voltage and reactive power sharing control based on non-iterative decoupled linearized power flow for islanded microgrids," *IEEE Access*, vol. 9, pp. 105242–105254, 2021.
- [35] J. Yang, N. Zhang, C. Kang, and Q. Xia, "A state-independent linear power flow model with accurate estimation of voltage magnitude," *IEEE Trans. Power Syst.*, vol. 32, no. 5, pp. 3607–3617, Sep. 2017.
- [36] H. R. Chamorro and N. L. Díaz, "Hierarchical power flow control in low voltage microgrids," in *Proc. North Amer. Power Symp. (NAPS)*, Sep. 2013, pp. 1–5.
- [37] S. Sumathi, L. A. Kumar, and P. Surekha, Solar PV and Wind Energy Conversion Systems. Cham, Switzerland: Springer, 2015.
- [38] S. Thakar, A. S. Vijay, and S. Doolla, "Effect of P-Q limits on microgrid reconfiguration: A capability curve perspective," *IEEE Trans. Sustain. Energy*, vol. 11, no. 3, pp. 2040–2048, Jul. 2020.
- [39] M. A. Hossain, H. R. Pota, S. Squartini, F. Zaman, and K. M. Muttaqi, "Energy management of community microgrids considering degradation cost of battery," *J. Energy Storage*, vol. 22, pp. 257–269, Apr. 2019.
- [40] A. Berrueta, J. Pascual, I. S. Martin, P. Sanchis, and A. Ursua, "Influence of the aging model of lithium-ion batteries on the management of PV self-consumption systems," in *Proc. IEEE Int. Conf. Environ. Electr. Eng. IEEE Ind. Commercial Power Syst. Eur. (EEEIC/ICPS Europe)*, Jun. 2018, pp. 1–5.
- [41] A. El Mejdoubi, H. Chaoui, H. Gualous, P. Van den Bossche, N. Omar, and J. Van Mierlo, "Lithium-ion batteries health prognosis considering aging conditions," *IEEE Trans. Power Electron.*, vol. 34, no. 7, pp. 6834–6844, Jul. 2019.
- [42] J. Schiffer, D. U. Sauer, H. Bindner, T. Cronin, P. Lundsager, and R. Kaiser, "Model prediction for ranking lead-acid batteries according to expected lifetime in renewable energy systems and autonomous power-supply systems," *J. Power Sources*, vol. 168, no. 1, pp. 66–78, May 2007.
- [43] S. Boyd, N. Parikh, E. Chu, B. Peleato, and J. Eckstein, "Distributed optimization and statistical learning via the alternating direction method of multipliers," *Found. Trends Mach. Learn.*, vol. 3, no. 1, pp. 1–122, 2010.
- [44] J. Bu, M. Fazel, and M. Mesbahi, "Accelerated consensus with linear rate of convergence," in *Proc. Annu. Amer. Control Conf. (ACC)*, Jun. 2018, pp. 4931–4936.
- [45] T. Brudigam, D. Prader, D. Wollherr, and M. Leibold, "Model predictive control with models of different granularity and a non-uniformly spaced prediction horizon," in *Proc. Amer. Control Conf. (ACC)*, May 2021, pp. 3876–3881.
- [46] B. Xu, A. Oudalov, A. Ulbig, G. Andersson, and D. S. Kirschen, "Modeling of lithium-ion battery degradation for cell life assessment," *IEEE Trans. Smart Grid*, vol. 9, no. 2, pp. 1131–1140, Mar. 2018.
- [47] C. Tautiva and A. Cadena, "Optimal placement of distributed generation on distribution networks," in *Proc. IEEE/PES Transmiss. Distrib. Conf. Expo., Latin Amer.*, Aug. 2008, pp. 1–5.
- [48] C. Voyant, G. Notton, J.-L. Duchaud, L. A. G. Gutiérrez, J. M. Bright, and D. Yang, "Benchmarks for solar radiation time series forecasting," *Renew. Energy*, vol. 191, pp. 747–762, May 2022.



**GERMAN OBANDO** received the B.S. degree in electronics from the Universidad de Nariño, Pasto, Colombia, in 2008, the M.S. degree in electronics engineering from the Universidad de los Andes, Bogotá, Colombia, in 2010, and the Ph.D. degree in electronics engineering from the Universidad de los Andes and the Ecole des Mines de Nantes, France, in 2016. He joined the School of Engineering, Science and Technology of Universidad del Rosario, Colombia, in 2017, where he is currently

an Assistant Professor. His current research interests include population dynamics, distributed control, and time delayed systems.

**HAROLD R. CHAMORRO** (Senior Member, IEEE) is currently with the KTH, Royal Institute of Technology, Stockholm, Sweden.



**ANDRÉS PANTOJA** received the B.S. degree in electronics engineering from Universidad Nacional, Manizales, Colombia, in 1999, and the M.S. and Ph.D. degrees in electronics engineering from the Universidad de los Andes, Bogotá, Colombia, in 2008 and 2012, respectively. In 2003, he joined the Departamento de Electrónica, Universidad de Nariño, Pasto, Colombia, where he is currently an Associate Professor and the Head of the Department of Electronics. His research inter-

ests include dynamic resource allocation, distributed generation, distributed control in smart grids and buildings, and coordination in large-scale systems.



**EDUARDO CAICEDO BRAVO** received the degree in electrical engineering from the Universidad del Valle, in 1985, the Master of Information Technology degree in manufacturing, in 1993, and the Doctor Engineering (Ph.D.) degree in industrial informatics from the Polytechnic University of Madrid-Spain, in 1996. He is currently a Professor with the Universidad del Valle and the Director of the Research Group Perception and Intelligent Systems (PSI). His research interests

include instrumentation electronics, intelligent systems, computational intelligence, and robotics.



**JOHN BARCO-JIMÉNEZ** received the B.S. degree in electronics from the Universidad de Nariño, Pasto, Colombia, in 2009, and the M.S. degree in electronics engineering from the Universidad de los Andes, Bogotá, Colombia, in 2013. He is currently pursuing the Ph.D. degree in engineering with an emphasis in electrical and electronic engineering with the Universidad del Valle, Cali, Colombia, and the Ph.D. degree in electrical power systems with the Universidad de Málaga,

Málaga, Spain. He works as a Professor and a Researcher with the Universidad CESMAG and the Universidad de Nariño, on electrical microgrids and automation and control systems.



**JOSÉ A. AGUADO** (Member, IEEE) was born in Málaga, Spain. He received the degree in electrical engineering and the Ph.D. degree from the University of Málaga, Málaga, in 1997 and 2001, respectively. He is currently a Full Professor and the Head of the Electrical Engineering Department, University of Málaga. He has led more than 40 publicly funded research and consulting projects on the operation and planning of smart grids and wireless power transfer.

....